

Article



Zero-Inflated Generalized Linear Mixed Models: A Better Way to Understand Data Relationships

Luiz Paulo Fávero ^{1,*,†}, Joseph F. Hair, Jr. ^{2,†}, Rafael de Freitas Souza ^{1,†} and Matheus Albergaria ^{3,†} and Talles V. Brugni ^{4,†}

- ¹ Business and Management Department, School of Economics, Business and Accounting, University of São Paulo—USP, São Paulo 05508-010, Brazil; fsrafael@usp.br
- ² Marketing & Quantitative Methods, Mitchell College of Business, University of South Alabama, Mobile, AL 36688, USA; jhair@southalabama.edu
- ³ Postgraduate Department of Business Management and Accounting, Fundação Escola de Comércio Álvares Penteado—FECAP, São Paulo 01005-010, Brazil; matheus.fecap@gmail.com
- ⁴ Finance and Accounting Department, FUCAPE Business School, Vitória 29075-505, Brazil; tallesbrugni@fucape.br
- Correspondence: lpfavero@usp.br; Tel.: +55-11-3091-6045
- + These authors contributed equally to this work.

Abstract: Our article explores an underused mathematical analytical methodology in the social sciences. In addition to describing the method and its advantages, we extend a previously reported application of mixed models in a well-known database about corruption in 149 countries. The dataset in the mentioned study included a reasonable amount of zeros (13.19%) in the outcome variable, which is typical of this type of research, as well as quite a bit of social sciences research. In our paper, present detailed guidelines regarding the estimation of models where the data for the outcome variable includes an excess number of zeros, and the dataset has a natural nested structure. We believe our research is not likely to reject the hypothesis favoring the adoption of mixed modeling and the inflation of zeros over the original simpler framework. Instead, our results demonstrate the importance of considering random effects at country levels and the zero-inflated nature of the outcome variable.

Keywords: zero-inflated models; count data models; multilevel models; mixed models; random coefficients models; hierarchical models; random effects; nested models; GLMM

1. Introduction

Count data models have increasingly been applied in management research, but the number of studies is still small and further development of this area is needed. As the name suggests, these models have an outcome variable that is a count variable, i.e., a quantitative variable that assumes non-negative and discrete values. Count data models also include several predictor variables, and the statistical objective is to identify and rank the predictor variables that maximize the ability to predict membership in the values of the outcome variable.

The study of Blevins et al. [1] deserves mention since the authors present an in-depth overview of count data models in management. Examples of applications involving count models are the number of times a business firm issues stocks, the number of registered patents over time, and the quantity of branches a company owns/operates. These counts model applications are usually analyzed using generalized linear modeling (GLM), as described by Almeida et al. [2] and O'Hara and Kotze [3], where estimations focus on understanding and explaining how expected counts of membership in the values of the outcome variable change as a function of predictor variables.

In a seminal work, Lambert [4] was bothered by the biases generated by count data models in the presence of a considerable amount of zeros in the outcome variable. Thus,



Citation: Fávero, L.P.; Hair, J.F., Jr.; Souza, R.d.F.; Albergaria, M.; Brugni, T.V. Zero-Inflated Generalized Linear Mixed Models: A Better Way to Understand Data Relationships. *Mathematics* **2021**, *9*, 1100. https:// doi.org/10.3390/math9101100

Academic Editors: Octavian Dospinescu and Juan Jose García-Machado

Received: 28 March 2021 Accepted: 30 April 2021 Published: 13 May 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). a proposal of a GLM estimation arises that combines a logistic component, in order to identify and better model the excess of zeros (the so-called structural zeros), with a count component which aims to model count data including the remainder of zeros probabilistically considered in the theoretical distributions of this type of estimation (the so-called sample zeros).

This type of GLM is known as a zero-inflated count model. In this type of count model, as already said, the researcher models the outcome variable as a count variable with an excess number of zeros. For instance, when analyzing a firm's number of patents, one may notice that there are observations when patents were not issued. In such situations, the researcher has two alternatives to obtain model solutions: (1) use methods to deal with the excessive zeros present in the data, or (2) estimate zero-inflated models. Several approaches are available to overcome the first point—excessive zeros in your data (i.e., methods aimed at dealing with outliers). But there are clear alternatives related to the second point—estimating zero-inflated models, since this second approach considers an outcome variable with excess zeros as part of the true data-generating process [5].

However, a limitation of traditional models such as GLM is they don't allow one to study individual heterogeneities, as well as differences among groups or contexts to which these individuals belong to, or for the specification of random components in different levels (such as groups) of the analysis [6,7]. In other words, traditional GLM methods disregard the natural nesting of the observations, based on the premise that such individuals do not share the same observational context [8], despite the fact that some students can be in the same school or some people share same cultural, social or religious aspects, for instance.

On the other hand, the so-called generalized linear mixed models (GLMM)-also called multilevel models, random coefficients models, nested models, or hierarchical models-act precisely to consider those limitations of the GLM techniques. GLMM estimates, therefore, take into account the existence of dependence among observations from the same group and, consequently, generate unbiased standard errors. Additionally, if the intention is to study if predictor group-level variables interact with individual-level variables, GLMM may be the most adequate estimation [9].

In this sense, the sudy published by Hall [10] deserves mention since it expands Lambert's [4] ideas to the field of GLMM estimations, considering not only the modeling of the structural and sample zeros, but also the modeling of the count component, taking into account the latent hierarchical groupings evidenced in the dataset.

With this in mind, the objective of this article is to present detailed guidelines regarding the estimation of models where the data for the outcome variable includes an excess number of zeros and the dataset has a natural nested structure. To do so, we use as a starting point one of the well-known models proposed by Fisman and Miguel [11] and compare it with a multilevel zero-inflated negative binomial model (ZINBM) that uses the same variables, but now taking into account the natural nesting of the observations which were originally in their dataset, but were not considered.

This article fills an important gap in methods development since zero-inflated generalized linear mixed models are not yet well explored in management research (according to the best of our knowledge), as discussed in the next sections. We illustrate the main steps related to the estimation of count data, zero-inflated, and zero-inflated multilevel models, and, at the same time, we present estimates of these distinct count models based on a dataset related to corruption practices in a public organization.

The paper is structured as follows: Section 2 discusses zero-inflated and multilevel models, and presents an overview of the articles related to these topics published in the Top 10 journals in the field of strategic management. Section 3 provides an overview on count data models, such as Poisson, negative binomial (NB) with overdispersion concept and zero-inflated. Section 4 presents the generalized linear mixed models (GLMM), with focus on the estimation of the parameters. In Sections 5 and 6, we offer an illustrative dataset for estimating GLMM through the package glmmTMB available at the Comprehensive R Archive

Network (CRAN; http://cran.us.r-project.org (accessed on 28 March 2021)). Section 7 presents our conclusions. Technical details on count data statistical distributions and zero-inflated models, and the R codes, are included in the Appendices A and B.

2. Background

Zero-inflated regression models are considered a combination of a model for count data with a model for binary data [4]. They are used to identify the reasons why a particular quantity of counts occurs regardless of the number of observed counts. Two types of zero-inflated models are typically estimated, being the first related to the zero-inflated Poisson (ZIP) model estimated from the combination of a Bernoulli distribution with a Poisson distribution, while the second is related to a zero-inflated negative binomial (ZINB) model estimated from the combination of a Bernoulli distribution with a Poisson-Gamma distribution. And one can choose between these two types taking into account existence of overdispersion in the data, i.e., analyzing if the variance of the outcome variable is statistically higher than its mean, as we will discuss in Section 3.

Research in social sciences has dedicated only limited time to the topic of zero-inflated models. A brief review of publications on the topic confirms this point. Table 1 shows the results of searches for terms related to both count models and zero-inflated models in the field of Social Sciences (scholar.google.com.br/ (accessed on 28 March 2021)). Specifically, the table displays each journal's ranking (column (2)), as well as the number of times the term "Zero-Inflated" is found in a search at each website (column (3)). Table 1 also contains the number of times the terms "Poisson", "Negative Binomial" and "Overdispersion" appear (columns (4), (5), and (6), respectively). The seventh column combines columns (4) + (5) - (6). Thus, the last column adds the number of times the terms "Poisson" and "Negative Binomial" (two of the most commonly used models in count data analysis) are found, and then subtracts the term "Overdispersion". The latter term is subtracted to avoid the problem of double counting similar citations, since some of the cited papers may apply both Poisson and NB models when data overdispersion is present. In general, therefore, the seventh column can be considered a proxy for the use of count-based models in social sciences research. The main goal of this search was to obtain initial evidence of the relative importance of zero-inflated models as compared to count-based models.

When analyzing the results displayed in the last column, one notices that most journals present values in the 5–12% range (the average ratio is 9.00%, while the median is 7.60%). At first, this informal evidence reinforces the notion that zero-inflated models are still underused in contemporary social sciences research, a result in line with previous studies [12].

The results in Table 1 demonstrate that, while there is considerable evidence of count models in social sciences research, there is very limited reporting of zero-inflated journal's aims and scope, as well as editorial policy, and a more detailed exploration of these empirical patterns appears justified.

It could also be argued that the existence of inflation of zeros in the outcome count variable is a rare event when compared to the non-existence of inflation of zeros. Although the reasoning may make sense for some fields of knowledge, in the field of social sciences, the excess of zeros in the studied phenomenon is not uncommon [13]. Interesting examples of phenomena whose distribution often contains inflation of zeros are traffic accidents [14], insurance claims [15], the arrests unleashed due to the occurrence of human trafficking crimes [16], the behavior of consumption and dependence on alcohol, cigarettes and other drugs by adolescents [17], unemployment issues and their impacts on mental health [18], social, demographic, and economic factors and their impacts on the current Sars-Cov-2 pandemic [19,20]. And the field considered in this paper should also be mentioned, since cultural aspects influencing practices of corruption have not yet been studied from the perspective of the zero-inflated models.

Blevins et al. [1] previously presented a detailed exposition of count models specifically in management research. The authors started their analysis by reviewing 11 years of management research using count-based outcome variables in 10 leading journals. They found that, while one of four papers in their sample used the basic Poisson regression model to estimate count outcome variables, alternative models might have been more appropriate in such occasions.

In fact, in many of the applications considered by the authors, we believe an alternative zero-inflated model might have provided a better fit for the data. In drawing this conclusion, the authors compared different count models estimated with previously published data and, at the same time, ran simulations based on different parameter values. The results also concluded that, in some cases, there was the possibility of significance level changes in some regressors, as well as sign changes.

In addition, we argue that the consideration of the natural nesting of the observations should also be taken into account in conjunction with the inflation of zeros for the count data models. This is an important qualification since GLM do not take into account a possible grouping of observations within a nested data structure. However, household residences belonging to the same neighborhood, or different students from the same school are often correlated, and these correlations can be accounted for in GLMM with the consideration of random effects [21,22] (As the authors note: "Despite the benefits of using these models, the use of zero-inflated models is only recently gaining traction in management research. Our review showed that only a small percentage of management scholars (11%) used such models" [1]. We view the results presented in Table 1 as robust evidence to these authors' claims.).

In short, in comparison with the classic GLM, GLMM estimations have the advantage of proposing a structure with different levels, and each of these levels is investigated by its own model. According to Woltman et al. [23], each level tries to capture the behavior of the variables and specifies how these variables are related to and influence other levels. As discussed by Courgeau [8], GLMM enables the researcher to relate observations and contexts, such as firms and countries, students and schools, or families and neighborhoods. If one decides to ignore these relationships, incorrect interpretations can arise.

The mixed designation in the GLMM term comes from the fact that predictor variables can be considered in both fixed and random effects components of the regression model. The estimated parameters of fixed effects indicate the relationship between predictor variables and the outcome variable (as well as in the context of GLM estimations), while the random effects component can be represented by the combination of error terms and predictor variables. Recent contributions on the benefits of mixed models are found in Reference [24–31]. All of these studies emphasized that multilevel models are a generalization of regression methods and, thus, can be applied for prediction and also for causal inference based on observational studies.

Mixed models, a term that also commonly appears in the literature as multilevel models, hierarchical models or random coefficients models have acquired considerable importance in the social sciences, and the publication of papers that consider these estimations have been increasingly more frequent, mainly due to studies that take into account the existence of nested data structures. Table 2 displays the search results for the terms "Multilevel Model", "Hierarchical Model", and "Random Coefficients Model" in Social Sciences (scholar.google.com.br/(accessed on 28 March 2021)). While the analysis shows a meaningful number of papers in top social sciences research journals that apply some type of mixed modeling, our research was unable to find in these same journals a single paper with zero-inflated estimations and, simultaneously, with a multilevel perspective, i.e., that takes into account the existence of a nested data structure.

In addition to what is shown in Tables 1 and 2, studies that estimate models taking into account, simultaneously, the inflation of zeros in the outcome variable and the multilevel structure of the dataset are quite rare. Based on our search, zero-inflated generalized linear mixed models are not yet very widely explored in social sciences.

Journal Name (1)	Google Scholar's Ranking (2)	"Zero -Inflated" (3)	"Poisson" (4)	"Negative Binomial" (5)	"Overdispersion" (6)	Count Models (7) = (4) + (5) - (6)	ZIM Relative Participation (8) = (3)/(7)
Journal of Business Ethics	1	4	36	22	6	52	7.69%
Computers Education	2	2	22	2	0	24	8.33%
American Journal of Public Health	3	32	549	151	85	615	5.20%
Research Policy	4	48	236	248	80	404	11.88%
Global Environmental Change	5	8	36	19	6	49	16.33%
Health Affairs	6	9	76	57	13	120	7.50%
Social Science Medicine	7	36	422	153	60	515	6.99%
New Media Society	8	0	13	15	4	24	0.00%
American Journal of Political Science	9	14	53	54	32	75	18.67%
Procedia-Social and Behavioral Sciences	10	9	100	27	6	121	7.44%

Table 1. Use of count models and zero-inflated models (ZIM) in the top 10 journals in the field of Social Sciences.

Notes: Authors' calculations, based on Google Scholar and journals' data. Search covers all years for all journals. Journals' ranking positions (column 1) were obtained from Google Scholar for the field of "Social Sciences". Searches in specific journals were made for the terms "Zero-Inflated" (column 3), "Poisson" (column 4), "Negative Binomial" (column 5), and "Overdispersion" (column 6). Column (7) was obtained by the following column operation: (4) + (5) - (6). The latter column is a proxy for the use of count models in social sciences research top journals. Finally, column (8) was obtained by the column operation: (3)/(7). This column represents the relative participation of zero-inflated models (ZIM) in published articles employing count models. Table completed on 19 March 2021.

Journal Name (1)	Google Scholar Ranking (2)	"Multilevel Model" (3)	"Hierarchical Model" (4)	"Random Coefficients Model" (5)	GLMM (6) = (3) + (4) + (5)
Iournal of Business Ethics	1	60	47	13	120
Computers & Education	2	62	27	0	88
American Journal of Public Health	3	59	20	2	81
Research Policy	4	35	22	6	63
Global Environmental Change	5	25	13	1	39
Health Affairs	6	2	7	0	9
Social Science Medicine	7	727	149	36	912
New Media Society	8	2	7	0	9
American Journal of Political Science	9	38	17	1	56
Procedia-Social and Behavioral Sciences	10	72	144	6	22

Table 2. Use of GLMM in the top 10 journals in the field of Social Sciences.

Notes: Authors' calculations, based on Google Scholar and journals' data. Search covers all years for all journals. Journals' ranking positions (column 1) were obtained from Google Scholar for the field of "Social Sciences". Searches in specific journals were made for the terms "Multilevel Model" (column 3), "Hierarchical Model" (column 4), and "Random Coefficients Model" (column 5). Column (6) brings the sum of columns (3), (4) and (5), i.e., represents a proxy for the total amount of papers that make use of mixed models in social sciences research top journals. Table completed on 19 March 2021.

3. Count Data and Zero-Inflated Models

Understanding the probability distributions and algebraic and econometric criteria for estimating parameters is crucial for the definition of predictive equations that try to capture, in a more reliable way, the real behavior of data.

Nelder and Wedderburn [32], in a relevant paper, presented and discussed through theoretical and conceptual points of view, several works that had previously estimated logistic models involving Bernoulli and binomial distributions, count data models involving Poisson distribution, and polynomial models involving Gamma distribution, among others. All these model estimations were combined in a group referred to as GLM [29]. As part of GLM discussions, therefore, count data regression models can be estimated for cases in which the the otcome variable is quantitative with discrete and non-negative values. Moreover, both Cameron and Trivedi [33] and Hilbe [34,35] note that these estimations are more adequate that traditional linear regression models, which can fail to take into account the presence of discrete and non-negative values in the outcome variables.

Wooldridge [36] indicates that a general regression model for count data can be described using Expression (1) as follows:

$$\widehat{Y}_{i} = \lambda_{i} = \exp(\beta_{0} + \beta_{1}X_{1i} + \beta_{2}X_{2i} + \dots + \beta_{k}X_{ki}),$$
(1)

where $\lambda_i = \exp(\beta_0 + \beta_1 X_{1i} + \beta_2 X_{2i} + ... + + \beta_k X_{ki})$ represents the expected number of occurrences or incidence rate ratio of the phenomenon studied for a given exposure; β_0 is the intercept; $\beta_j (j = 1, 2, 3, ..., k)$ are the coefficients estimated for each predictor variable X_i ; k is the number of predictor variables used in the model; and i indicates each observation of a given sample.

While Poisson models assume the existence of equidispersion of the count data for the variable of interest, i.e., $\mu_i = E(Y_i) = Var(Y_i) = \lambda_i$, the estimation of a NB model assumes overdispersion of the outcome variable, conditional to predictor variables Hilbe [35], i.e., $\mu_i = E(Y_i) < Var(Y_i) = \lambda_i$. Expressions (2) and (3) represent the mean and the variance, respectively, for a negative binomial model.

$$E(Y_i) = u_i, \tag{2}$$

$$Var(Y_i) = u_i + \phi u_i^2, \tag{3}$$

where $\phi = \frac{1}{\psi}$ and $u_i = \exp(\beta_0 + \beta_1 X_{1i} + \beta_2 X_{2i} + \dots + \beta_k X_{ki})$.

As discussed by Cameron and Trivedi [37], the parameter ϕ in expression (3) represents overdispersion in the count data. According to Fávero et al. [38], in the cases where $\phi \rightarrow 0$, equidispersion would be detected in the dataset used, indicating that a Poisson regression model should be used. In addition, according to the authors, for the case in which ϕ is statistically greater than zero, the phenomenon of overdispersion would occur, which suggests the use of the NB model.

Although the most commonly used model for the analysis of count data is the Poisson model, by definition, its distribution has a single free parameter, λ , thus not allowing the variance to be fitted to the mean [33,39], as previously explained for the case of $\lambda = Var(Y)$. Thus, in the presence of overdispersion, a NB model can provide a better fit for the count data, in which the mean of the Poisson distribution can be used as a random variable that follows a Gamma distribution, together with an additional free parameter ϕ [40].

Additionally, it is common that some count data variables show a great amount of values equal to zero, which represents a fact that can generate biases in the parameters estimated through traditional Poisson or NB regression models, since these models do not take into account the heightened presence of zero counts. In these cases, the consideration of zero-inflated regression models makes sense [5].

In this perspective, according to Lambert [4], zero-inflated regression models are used for investigating the reasons why a certain number of counts occur, including zeros, and the reasons why this quantity happens, regardless of the number of observed counts. As stated in Ngatchou-Wandji and Paris [41], zero-inflated regression models can be defined as follows:

$$p(Y = y|\omega) = \omega\delta_0(y) + (1 - \omega)f(y), \tag{4}$$

where *Y* is the count variable, ω is the proportion of the excess of zeros, $\delta_0(y) = 1$ if y = 0 and $\delta_0(y) = 0$, otherwise; and f(y) is the density of a count distribution, such as Poisson or NB.

In this sense, ZIP models can be estimated through the consideration, simultaneously, of Bernoulli and Poisson distributions. On the other hand, ZINB models are estimated through the simultaneous consideration of Bernoulli and Poisson-Gamma distributions. And one can choose the best estimation depending on the existence of overdispersion in the data [13]. In other words, this decision can be made taking into account the statistical significance of the inverse of the Gamma distribution parameter. And the definition of the existence or not of an excessive amount of zeros in the outcome variable is determined through the analysis of the outputs of the Vuong Test [42]. Table 3 shows the relationship between count data models, overdispersion, and inflation of zeros.

Table 3. Count data regression models, overdispersion, and inflation of zeros in the outcome variable.

Varification		Count Data	Regression Mode	1
vernication	Poisson	Negative Binomial (NB)	Zero-Inflated Poisson (ZIP)	Zero-Inflated Negative Binomial (ZINB)
Overdispersion in Outcome Variable	No	Yes	No	Yes
Inflation of Zeros in Outcome Variable	No	No	Yes	Yes

Notes: Extracted from Fávero and Belfiore [29].

According to Table 3, while ZIP and ZINB regression models are more appropriate in the presence of an excessive amount of zeros in the outcome variable, the use of the latter is further recommended if there is overdispersion in the data. Technical details on statistical distributions, as well as on count data and zero-inflated regression models, are included in Appendix A and in the supplementary material.

4. Generalized Linear Mixed Models (GLMM)

Multilevel models have become quite important in several fields of knowledge, including management. The main reason is that some research constructs are taking into account the existence of nested, or mixed, data structures, in which certain variables vary among groups, or contexts, but do not vary among observations from the same context [21,43]. In addition, computational developments that have increased processing capacities allow the estimation, increasingly, of different types of multilevel models [29,44,45].

4.1. The Two-Level Model

According to Fávero [46], there are many situations where data are disposed within a mixed, or nested structure, and the hierarchies refer to fact that observations from the same groups, or contexts, share aspects, or characteristics, that represent a kind of homogeneity.

In this perspective, a two-level zero-inflated count data mixed model, in this sense, can be specified, where the first level refers to observations i(i = 1, ..., n), nested in two-level units j(j = 1, ..., J), as follows:

$$\ln\left(\frac{p_{logit_{ij}}}{1-p_{logit_{ij}}}\right) = \mathbf{Z}_{ij}^{T}\boldsymbol{\delta}, \text{ being } \boldsymbol{\delta} = \mathbf{G}_{j}^{T}\boldsymbol{\alpha}, \tag{5}$$

$$\ln(\lambda_{ij}) = \mathbf{X}_{ij}^T \boldsymbol{\pi} + \boldsymbol{\nu}_j, \text{ being } \boldsymbol{\pi} = \mathbf{H}_j^T \boldsymbol{\beta}, \tag{6}$$

where matrices of the predictor variables Z and X, of levels *i* (level 1) and *j* (level 2), appearing in the respective logistic (Bernoulli) and Poisson or Poisson-Gamma components

are not necessarily the same (equal conditions hold for matrices of the variables G_j and H_j at level two), and δ , π , α , and β are the respective matrices of the regression parameters, i.e., δ can be interpreted in terms of the proportion of inflation of zeros, π is related to the mean response in the count data part, and α and β correspond to differences among level two context in structural and sample zeros, respectively, due to the behavior of predictor variables in level two (G_j and H_j). Following Lee et al. [47] and Fávero et al. [40], δ_j represents the random variations at second level, which means that heterogeneity among higher levels of analysis (groups, for instance) and between individuals is allowed through the random effects δ , with variance equal to $\sigma^2 v_j$.

Despite the fact that Rabe-Hesketh et al. [48] discuss and estimate multilevel models taking into account discrete outcome variables, papers that consider zero-inflated count data regression models with random effects in the literature are still quite limited.

4.2. Model Estimation and Materials

As discussed by Lee et al. [47], one can estimate zero-inflated generalized linear mixed models considering the restricted maximum likelihood approach. The definition the GLMM likelihood function requires the specification of the log-likelihood function of the fixed component (first term), as well as the logarithm of the probability density function of the random effects (second term), which allows one to specify more complex correlation structures in the variance components.

Lee et al. [47] present the log-likelihood function for zero-inflated count data mixed models. The first term, LL_1 , is given by Expressions (A14) or (A19) presented in Appendix A and is related to ZIP and ZINB estimations, respectively, depending on the existence of overdispersion on data. The second term is given by:

$$LL_{2} = -\frac{1}{2} [n \ln(2\pi\sigma_{\nu}^{2}) + \sigma_{\nu}^{-2}\nu^{T}\nu].$$
⁽⁷⁾

Random effects v allow the existence of heterogeneity among clusters and also among individuals. Estimation proceeds by maximizing LL_1 (A14) or (A19) with the consequent update of the values of the variance components through the estimation of a restricted maximum likelihood (REML) function from LL2 [46]. Thus, $LL_1 + LL_2$ generates the final LL for a zero-inflated multilevel model. In the next section, we will estimate and present the parameters of one of the original models proposed by Fisman and Miguel [11], the NB estimation, which the authors called Model 3, including its analogous model that considers the inflation of zeros (ZINB estimation), as well as the model that considers the inflation of zeros under the multilevel perspective (ZINBM estimation).

With that said, first, one can study the statistical significance of the excessive amount of zeros in the outcome variable *Y* through the application of the Vuong test. Subsequently, the study can be conducted taking into account an eventual hierarchical structure of the observations in the dataset of Fisman and Miguel [11].

Finally, the log-likelihood values of the estimated models were compared to enable one to verify the best suitability for the considered data, and pointing out the biases generated by not considering the inflation of zeros and/or the nested structure of the data.

In this paper, all the estimations are obtained through the software R version 4.0.4, using the package MASS for the NB models, the package pscl for the ZINB model, and the package glmmTMB for the ZINBM model.

5. Culture and Corruption among United Nations' Diplomats: An Applied Example

Corruption is an important matter. For instance, when analyzing the effect of corrupt practices on economic outcomes for a cross-section of countries, Mauro [49] finds that corruption lowers private investment and long-run growth. In fact, there is an evergrowing literature aimed at studying and measuring corruption, both in individual and aggregate levels [50–52]. In this section, we present an application related to zero-inflated generalized linear mixed models for the study of corruption practices in organizations. We used the paper's code in Stata and respective datasets available at Edward Miguel's website, despite the fact that Fisman and Miguel also made available other versions. We first made extensive use of the materials available there, such as the authors' data codebook, for instance (The original datasets and code are available at the following links: https://dataverse.harvard.edu/dataset.xhtml?persistentId=doi:10.7910/DVN/28059 (accessed on 28 March 2021) and http://emiguel.econ.berkeley.edu/research/corruptionnorms-and-legal-enforcement-evidence-from-diplomatic-parking-tickets (accessed on 28 March 2021). As an additional note, we did not get in touch with the authors in order to use their public dataset.).

Fisman and Miguel [11] presented an original approach to study corruption in a sample of countries. The authors explore based on a unique point of view: the parking behavior of United Nations' (UN) diplomats. They analyzed parking violations from worldwide diplomats during the 1997–2005 period. Given that, since in part of the considered period diplomatic immunity protects UN diplomats from parking enforcement actions, while in the subsequent period diplomatic immunity for parking enforcement actions no longer occurs, the authors argue that diplomats' behavior regarding parking tickets can be considered a good proxy for cultural aspects which can determine corrupt behavior.

Based on this data, the authors built a "revealed preference" measure of corruption among officials who work for governments of 149 countries. This measure has the advantage of being based on diplomats' observed actions, which is its main feature, given the well documented difficulties in measuring corruption worldwide [53] (The authors also built a ranking of countries based on parking violations. In the first positions of the ranking come Kuwait, Egypt and Chad (as more corrupt countries), while Japan, Norway, and Sweden are among the least corrupt in the sample [11].).

Fisman and Miguel [11] bring an interesting discussion emphasizing that parking violation corruption is strongly (and positively) correlated with some country corruption measures, such as those defined by the World Bank surveys, for instance. In general terms, this result suggests that home country corruption norms may serve as a predictor for the propensity to behave corruptly among government officials. Specifically, the data reveals that diplomats from high-corruption countries (such as Nigeria, for instance), on average, tend to behave remarkably bad in situations in which there are no legal consequences involved, while the opposite is true for diplomats from low-corruption countries (such as Norway), whose diplomats do not tend to commit more parking violations.

We begin by replicating a graphical pattern contained in the data. Figure 1 shows the scatterplot relating unpaid parking violations (vertical axis, in natural logarithmic scale) and country corruption measures (horizontal axis), considering the period of non-existence, as well as the period of existence of legal consequences for unpaid parking violations by diplomats (in Figure 1, the label 'no' means no legal consequences, while the label 'yes' means the opposite). The plot suggests a positive (and possibly nonlinear) relationship between diplomats' traffic violations and each country's corruption measures.



Figure 1. Parking violations and country corruption index from Fisman and Miguel [11] dataset.

As a second step, Figure 2 presents a histogram containing unpaid diplomats' parking violations, i.e., the outcome variable on Fisman and Miguel's Model 3, called by the authors *violations_all*. We construct this figure to provide a visual representation of zero values contained in the outcome variable, as well as to emphasize the likely overdispersion present in the outcome variable.



Figure 2. Histogram of the outcome variable (*violations_all*).

The graphical patterns described in Figure 2 shows the occurrence of zero values in 13.19% of the sample containing the outcome variable. At first, there is still no way to affirm the need of estimation of zero-inflated models based only on Figure 2, and more formal tests are needed before reaching stronger conclusions. Table 4 presents the variables used in Fisman and Miguel's Model 3, which were considered in the replication model under the multilevel perspective with inflation of zeros. For more information on the variables, see Fisman and Miguel [11].

Table 4. Description	of the variables	considered ir	n the Fisman	and Miguel's Model 3.

Variable	Description
wbcode	Polychotomic categorical variable that indicates the acronyms of the countries in the dataset.
region	Polychotomic variable that points to a region of the globe to which each country in the dataset be- longs: Africa, Asia, Europe, Middle East, North America, Oceania, and South America.
violations_all	Discrete variable that measures the amount of unpaid parking violations by diplomats in a given country. It is the outcome variable of the models.
corruption	Continuous variable that indicates the level of corruption in each country. The greater the polari- ty, the greater the country's corruption.
post	Dichotomous variable that indicates the moments that consider the inexistence and existence of le- gal consequences to unpaid parking violations by diplomats-'no' and 'yes' indicators, respective- ly.
staff	Discrete variable that measures the number of diplomats from a given country.
lgdppcus	Continuous variable that indicates the natural logarithm of GDP per capita of each country.

Table 5 presents the descriptive statistics of the quantitative variables, as well as the frequency tables of the qualitative variables explained in Table 4, which, in fact, were considered in the so-called Model 3 proposed by Fisman and Miguel [11].

Table 5. Univariate descriptive statistics of the quantitative variables and frequency table of the qualitative variables that were considered in the models.

Violation_all	Corruption	Post	Staff	Lgdppcus	Region
Min.: 0.0 1st Q.: 2.0 Median: 18.5 3rd Q.: 253.8 Max: 16,751.0 Mean: 494.6	Min.: -2.583 1st Q.: -0.415 Median: 0.327 3rd Q.: 0.720 Max: 1.583 Mean: 0.014	Freq. of 'no': 149 Freq. of 'yes': 149	Min.: 2.000 1st Q.: 6.000 Median: 9.000 3rd Q.: 14.000 Max: 86.000 Mean: 11.810	Min.: 4.559 1st Q.: 6.036 Median: 7.266 3rd Q.: 8.543 Max: 10.505 Mean: 7.394	Africa: 92 Asia: 50 Europe: 70 Middle East: 30 North America: 12 Oceania: 8 South America: 36

Note 1: Q. stands for quartile. Note 2: Freq. stands for Frequency. Note 3: North America data includes Caribbean data.

In terms of econometric methodology, Fisman and Miguel (2007) estimated a count data NB regression model. In Table 6, in addition to the original NB model proposed by Fisman and Miguel [11], we present a ZINB estimation (necessary for the application of the Vuong test), and a ZINBM estimation using exactly the same variables as the initial NB model.

Table 6. Estimations of negative binomial model [11], and correspondent zero-inflated and zero-inflated mixed model.

	NB–Model 3 Fisman and Miguel [11]	ZINB	ZINBM
	Fixed Effects		
intercent	2.871 ***	3.115 ***	2.732 **
intercept	(0.987)	(0.909)	-1,150
comunitor	0.575 ***	0.475 ***	0.561 ***
corruption	(0.162)	(0.159)	(0.215)
nost	-4.215 ***	-4.284 ***	-4.265 ***
post	(0.175)	0.153	(0.136)
staff	0.051 ***	0.048 ***	0.040 ***
Stall	(0.008)	(0.011)	(0.011)
ladonque	0.091	0.110	0.105
iguppeus	(0.116)	(0.112)	(0.129)
Africa	2.865 ***	2.657 ***	2.885 ***
Annca	(0.504)	(0.466)	(0.642)
Acia	1.985 ***	1.771 ***	2.133 ***
Asia	(0.518)	(0.484)	(0.668)
Furopo	2.237 ***	1.991 ***	2.020 ***
Lutope	(0.510)	(0.500)	(0.656)
Middle Fact	3.229 ***	3.084 ***	3.045 ***
Middle East	(0.544)	(0.533)	(0.707)

	NB–Model 3 Fisman and Miguel [11]	ZINB	ZINBM
Occaria	1.508 **	1.399 **	1.832 **
Oceania	(0.739)	(0.679)	(0.922)
South Amorica	1.667 ***	1.453 ***	1.671 **
South America	(0.533)	(0.515)	(0.684)
	Logistic Component		
intercent		-2.432 ***	-2.331 ***
intercept	-	(0.298)	(0.260)
comunitor		-0.695 ***	-0.564 ***
corruption	-	(0.204)	(0.212)
	Random Effects		
$Var(v_{0j})$ -intercept	-	-	0.502
$Var(v_{1j})$ -slope (corruption)	-	-	0.296
Observations	298	298	298
$\frac{1}{1}$	-1547.691	-1526.838	-1515.049
Log-likelihood (LL)	(df = 12)	(df = 14)	(df = 17)

Table 6. Cont.

Note: Standard errors are in parentheses. The omitted region category is North America/Caribbean. ** Statistically significantly different from zero at 95 percent confidence. *** Statistically significantly different from zero at 99 percent confidence.

6. Results and Discussions

The confirmation of overdispersion of the outcome variable, conditional to predictor variables, comes from the Cameron and Trivedi test [37], whose result is in Figure 3. In Appendix A, we present further details on the diagnosis of overdispersion.

```
Overdispersion Test - Cameron & Trivedi (1990)
data: parking_data
Lambda t test score: = 7.4955, p-value = 7.644e-13
alternative hypothesis: overdispersion if lambda p-value is
less than or equal to the stipulated significance level
```

Figure 3. Result of the Cameron and Trivedi overdispersion test.

The verification of the existence of excess of zeros in the outcome variable unpaid parking violations (*violations_all*) can be elaborated in sequence, through the Vuong test, which compares likelihood functions between a GLM estimation, such as the NB, with the respective estimation with eventually an inflation of zeros. Table 7 shows the results for the Vuong test when comparing the NB regression model with the corresponding ZINB model.

Table 7. Results of the Vuong test-ZI negative binomial x negative binomial.

	Vuong z-Statistic	<i>p</i> -Value
Raw	2.953160	0.0015727
AIC-corrected	2.669934	0.0037933
BIC-corrected	2.146377	0.0159215

Notes: Vuong non-nested hypothesis test-statistic: test-statistic is asymptotically distributed N (0,1) under the null hypothesis that the models are indistinguishable.

Since the Vuong test might be biased towards supporting models with zero-inflation components, according to Desmarais and Harden [54], we implemented Desmarais and Harden's correction to verify if the former results are somehow affected. While the Vuong test statistic is z = 2.95, the AIC and BIC corrected statistics are z = 2.67 and z = 2.15,

respectively, or rather, all present *p*-values < 0.05. These results indicate the better adequacy of the zero-inflated negative binomial regression model, in comparison with the traditional GLM negative binomial regression model estimated by Fisman and Miguel [11].

Such a diagnosis, which points out the better suitability of the ZINB estimation, by itself, would already suggest that the original NB estimation present biases by not considering the excess of zeros in the outcome variable. Furthermore, the NB estimation fails to identify that the slope of the variable *corruption*, in the ZINB logistic component, indicates that an increase in one unit leads to a 50.01% decrease, on average, in the chance of structural zeros, *ceteris paribus*, since $\exp(-0.695) = 0.499$. In other words, the variable *corruption* has an even greater preponderance in the absence of unpaid parking violations by diplomats in the considered countries.

This fact leads to a second important bias: the NB model overestimates almost all parameters of the other variables that were shown to be statistically significant, when compared to the count component of the ZINB estimation, although there is no inversion of signs-the exception occurs for the variable *post*.

When the original NB model is compared to the ZINBM estimation, which takes into account the nested structure of data among countries, the biases are more striking. The first point to note is that the consideration of these groupings leads to a smoothing of the slope of the variable *corruption* in the logistic component in comparison with the ZINB model.

The value of the slope parameter of the ZINB logistic component is -0.695; for the ZINBM estimation, -0.564. In other words, when considering the nested structure among countries, the chance of existence of structural zeros in the dataset is reduced, on average, by 43.10% when increasing the variable *corruption* in one unit, *ceteris paribus*, since $\exp(-0.564) = 0.569$. It is also interesting to note that the NB model cannot even capture the ratio of the inflation of zeros in the dataset and, even if the ZINB model succeeds, it reasonably overestimates the studied situation by neglecting the natural nesting in the data.

When considering the existing nested structure in the dataset of Fisman and Miguel [11], we realize that there is an overestimation of the parameters of the NB model, regarding both the general intercept and the slopes of the predictor variables *corruption*, *staff*, *Europe*, and *Middle East*, when compared to the ZINBM model. On the other hand, the NB estimation underestimates the parameters of the other variables when compared to the ZINBM estimation. There was also no inversion of signals.

Thus, in the NB model, the dummy variable that identifies the Oceania region has a coefficient equal to 1.508, against 1.832 for the ZINBM model. Thus, when adopting the North America region as the reference region, the NB model predicts, on average, an increase in the chance of occurrence of the outcome variable in 351.77%, if the diplomat is from Oceania, keeping the other conditions constant. The ZINBM model, on the other hand, intensifies this situation, since the increase in the chance of occurrence of the outcome variable, on average, is 524.64% if the diplomat is from Oceania, taking the region *North America* as a reference, *ceteris paribus*. In short, the NB model underestimates the occurrence of unpaid parking violations among diplomats from Oceania region by 67.05% when compared to the ZINBM estimation. Using this same approach, when compared to the ZINBM model underestimates the impact of the Asia region by 15.95%.

The parameter related to the Europe region also draws attention. The NB model overestimates the chance of unpaid parking violations among diplomats, on average, by 27.94%, *ceteris paribus*, when compared to the ZINBM model and being North America as the reference region. Under the same perspective, for the Middle East region, the NB estimate overestimates the impact by 20.20%.

Behind a functional form that is better adherent to the data of Fisman and Miguel [11], the ZINBM estimation has the ability to estimate random effects for the groups that characterizes the hierarchical levels—countries, in this case. Thus, the proposed model presents different error terms related to intercepts and slopes for each country.

Equation (8) presents the NB model proposed by Fisman and Miguel [11]; Equation (9) brings the result of the ZINB estimation, showing both the logistic and the count compo-

$$violations_all_{NB_i} = 2.871 + 0.575.corruption_i - 4.215.post_i + 0.051.staff_i + 0.091.lgdppcus_i + 2.865.Africa_i + 1.985.Asia_i + 2.237.Europe_i + 3.229.MiddleEast_i + 0.091.lgdppcus_i + 2.865.Africa_i + 1.985.Asia_i + 2.237.Europe_i + 3.229.MiddleEast_i + 0.091.lgdppcus_i + 2.865.Africa_i + 1.985.Asia_i + 2.237.Europe_i + 3.229.MiddleEast_i + 0.091.lgdppcus_i + 0.091.lgdpcus_i + 0.091.lgdpcus_i + 0.09$$

 $1.508.Oceania_i + 1.667.SouthAmerica_i$, (8)

$$\widehat{violations}_{all_{ZINB_{i}}} = \left\{ 1 - \frac{1}{1 + \exp[-(-2.432 - 0.695.corruption_{i})]} \right\} \cdot \exp[3.115 + 0.475.corruption_{i} - 4.284.post_{i} + 0.048.staff_{i} + 0.110.lgdppcus_{i} + 2.657.Africa_{i} + 1.771.Asia_{i} + 1.991.Europe_{i} + 3.084.MiddleEast_{i} + 1.399.Oceania_{i} + 1.453.SouthAmerica_{i}], (9)$$

$$\widehat{violations_all_{ZINBM_{ij}}} = \left\{ 1 - \frac{1}{1 + \exp[-(-2.3313 - 0.5643.corruption_{ij})]} \right\} \cdot \exp[2.732 + 0.561.corruption_{ij} - 4.265.post_{ij} + 0.040.staff_{ij} + 0.105.lgdppcus_{ij} + 2.885.Africa_{ij} + 2.133.Asia_{ij} + 2.020.Europe_{ij} + 3.045.MiddleEast_{ij} + 1.832.Oceania_{ij} + 1.671.SouthAmerica_{ii} + v_{0i} + v_{1i}.corruption_{ii}]. (10)$$

Looking again at Table 6, it is interesting to note that the increase in *LL* between the models could be said to be modest, but in all cases, according to a likelihood ratio test, it is statistically significant at 1%. More than that, when analyzing, for example, the fitted values of the ZINBM model considering the predictor variable *staff*, which has not yet been explored, it is difficult to advocate in favor of the NB model, as shown in Figure 4.



Figure 4. Fitted values of NB, ZINB, and ZINBM estimations, considering the predictor variable staff.

According to Figure 4, it is evident that the NB estimation is unable to adequately capture the inflation of zeros present in the dataset. It is also interesting to note that, for the ZINB estimation, even though it captures the excess of zeros better than the NB

model, there is no ability to adequately segregate, due to natural nesting (countries), the occurrence of structural zeros. In fact, the ZINBM estimation is superior to all others for the case studied.

7. Final Considerations

After exploring different approaches to a well-known NB model, we propose to divide our final considerations into three main parts: theoretical aspects, managerial implications and future research in count data multilevel modeling.

7.1. Theoretical Aspects

In this article, we presented guidelines for the estimation of zero-inflated generalized linear mixed models. That is, models where the outcome variable presents an excess number of zeros and the dataset has a structure where observations are correlated in ways that require random effects.

As an illustration of the importance of zero-inflated generalized linear mixed models, we presented an application based on a dataset related to corruption practices among UN's diplomats [11]. We emphasize that, in no way, we tried to diminish the merit of the study and the authors' findings. Furthermore, at the time of the study, multilevel models were rare and computational capabilities represented a considerable constraint. Our main message is about the superiority of GLMM models, in relation to GLM, when the natural nesting of observations is available to researchers.

We were able to replicate the original study's main results, confirming their authors' claims regarding the role of cultural norms for corrupt behavior. We found a strong effect of corruption norms: diplomats from high-corruption countries (on the basis of existing survey-based indices) accumulated significantly more unpaid parking violations, i.e., cultural norms can be seen particularly as an important determinant of corruption (Table 6 and Equations (8)–(10)).

We also extended their original analysis by estimating zero-inflated generalized linear mixed models. At one hand, the reported results confirm Fisman and Miguel's original findings, in terms of estimated signs and statistical significance (Table 6). On the other hand, the results allow us to discuss the importance of considering different zero-generating processes for outcome variables, also taking into account the nature and the structure of the considered dataset. In this sense, we are cannot reject the hypothesis favoring the use of ZINBM model over the traditional GLM models.

7.2. Managerial Implications

Management research still uses count-based models with parsimony [1,12]. Although there has been an increase in the use of such models, there is still considerable room for improvement, given the many opportunities related to interesting themes, such as patents, product innovations, and issue of shares, for instance. In fact, even when studying the influence of social factors over behavior, such as social ties, researchers might benefit from using count models [13].

While we believe the results presented here provide additional evidence supporting the use of count-based models, we emphasize the importance of zero-inflated generalized linear mixed models for specific applications. In a broader sense, these results are important for emphasizing potential uses of this class of models in distinct areas of management research. We agree with Blevins et al. [1], who discuss that there is still room for improvement in terms of existing research practices. Specifically, both researchers and practitioners could obtain substantial gains and insight by incorporating in their toolkit's models with excess zeros in the outcome variable in a multilevel perspective.

By presenting this guideline and applied example related to zero-inflated linear mixed models, we hope to see more studies of this kind in the near future, as well as to stimulate the development of new ways to analyze and interpret quantitative data in management research, as originally proposed by other authors [55,56]. By the way, Dale and Sirchencko [57] bring an important example in this regard.

In addition, we believe that datasets from organizational research sometimes do not explicitly represent contexts that can be defined as levels of analysis in multilevel modeling. Regardless, researchers and practitioners can create clusters from the observations' own data, allowing the establishment of latent higher-level random effects (in this case, at the level of the cluster itself), even if there are no variables corresponding to these clusters. Due to the fact that cluster analysis is an unsupervised exploratory technique, the predictive power of the estimation will be lost, but even so the model can be more adequately adjusted for diagnostic purposes regarding the behavior of the existing data.

7.3. Future Research in Count Data Multilevel Modeling

In our research, we did not find studies that estimate zero-inflated multilevel models using longitudinal data, in which the temporal dimension represents level 1 of the analysis. This can be an interesting prospect for future research.

Another relevant point is that we did not discuss endogeneity problems arising from the fact that entities such as individuals, organizations or countries can vary systematically from one another. As discussed by Antonakis et al. [58], researchers typically estimate multilevel models assuming the random effects are uncorrelated with the regressors. But this problem can be avoided by including cluster means of level 1 explanatory variables as controls.

From our research search, we did not find works that discuss these points and problems in estimations of Count Data Multilevel Models with inflation of zeros.

Supplementary Materials: The following are available online at https://www.mdpi.com/article/10 .3390/math9101100/s1.

Author Contributions: Conceptualization, L.P.F., J.F.H.J., R.d.F.S. and M.A.; Formal analysis, L.P.F., J.F.H.J. and R.d.F.S.; Methodology, L.P.F., J.F.H.J., R.d.F.S., M.A. and T.V.B.; Writing—original draft, L.P.F., J.F.H.J., R.d.F.S. and M.A.; Writing—review & editing, L.P.F., J.F.H.J., R.d.F.S. and T.V.B. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the School of Economics, Business and Accounting at the University of São Paulo—USP.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: The original datasets and code are available at the following links: https://dataverse.harvard.edu/dataset.xhtml?persistentId=doi:10.7910/DVN/28059 (accessed on 28 March 2021) and http://emiguel.econ.berkeley.edu/research/corruption-norms-and-legal-enforcement-evidence-from-diplomatic-parking-tickets (accessed on 28 March 2021). As an additional note, we did not get in touch with the authors in order to use their public dataset.

Conflicts of Interest: The authors declare no conflict of interest.

Sample Availability: The scripts for establishing the images and models of this study are available as supplementary materials.

Abbreviations

The following abbreviations are used in this manuscript:

- GLLM Generalized Linear Mixed Models
- GLM Generalized Linear Models
- LL Log-Likelihood Function

NB	Negative Binomial
UN	United Nations
ZINB	Zero-Inflated Negative Binomial
ZINBM	Multilevel Zero-Inflated Negative Binomial
ZIP	Zero-Inflated Poisson

Appendix A

Appendix A.1. Poisson Regression Models

According to Cameron and Trivedi [33], in general Poisson regression models can be applied when the distribution of the occurrence of a given phenomenon being studied follows a Poisson distribution, as shown in Figure A1, where *p* represents, for a determined observation i(i = 1, 2, ..., n), the probability of occurrence of a specific count for a determined exposition *m* (such as geographic region or period of time, for instance), as shown in Expression (A1).



Figure A1. The Poisson distribution.

$$p(Y_i = m) = \frac{\exp(-\lambda_i).\lambda_i^m}{m!}$$
, being $m = 0, 1, 2, ...$ (A1)

Klakattawi et al. [59] argue that Poisson models assume the existence of equidispersion of the count data for the variable of interest; i.e., $\mu_i = E(Y_i) = Var(Y_i) = \lambda_i$, in accordance with Expressions (A2) and (A3), as follows:

$$E(Y_i) = \sum_{m=0}^{\infty} m \frac{\exp(-\lambda_i) \cdot \lambda_i^m}{m!} = \lambda \sum_{m=1}^{\infty} \frac{\exp(-\lambda_i) \cdot \lambda_i^{m-1}}{(m-1)!} = \lambda$$
(A2)

$$Var(Y_i) = \sum_{m=0}^{\infty} \frac{\exp(-\lambda_i)\lambda_i^m}{m!} (m-\lambda)^2 = \sum_{m=0}^{\infty} \frac{\exp(-\lambda_i).\lambda_i^m}{m!} (m^2 - 2m\lambda + \lambda^2) = \lambda^2 \sum_{m=2}^{\infty} \frac{\exp(-\lambda_i).\lambda_i^{m-2}}{(m-2)!} + \lambda \sum_{m=1}^{\infty} \frac{\exp(-\lambda_i).\lambda_i^{m-1}}{(m-1)!} - \lambda^2 = \lambda \quad (A3)$$

The coefficients of a Poisson regression are estimated by the likelihood function given by Expression (A4) as follows:

$$L = \prod_{i=1}^{n} \frac{\exp(-\lambda_i).(\lambda_i)^{Y_i}}{Y_i!}$$
(A4)

from which the logarithm of the likelihood function given by Expression (A5) is derived.

$$LL = \sum_{i=1}^{n} [-\lambda_i + (Y_i) \ln(\lambda_i) - \ln(Y_i!)]$$
(A5)

which should be iterated until it reaches its maximum value.

Appendix A.2. Negative Binomial Regression Models and Overdispersion

The estimation of a NB model is directly linked to the presence of overdispersion in the count dataset [60], whose distribution is illustrated in Figure A2.



Figure A2. The negative binomial distribution.

In Figure A2, p is the likelihood of a random portion of the number of occurrences of the outcome variable Y_i in exposure m, as given by Expression (A6):

$$p(Y_i = m) = \begin{pmatrix} m + \psi - 1 \\ \psi - 1 \end{pmatrix} \left(\frac{\psi}{u_i + \psi}\right)^{\psi} \left(\frac{u_i}{u_i + \psi}\right)^m$$
(A6)

where $\psi(\psi > 0)$ is the shape parameter. As discussed, a NB model assumes overdispersion of the outcome variable, conditional to predictor variables ($\mu_i = E(Y_i) < Var(Y_i) = \lambda_i$, being $E(Y_i) = u_i$ and $Var(Y_i) = u_i + \phi u_i^2$, where $\phi = \frac{1}{\psi}$ and $u_i = \beta_0 + \beta_1 X_{1i} + \beta_2 X_{2i} + ... + \beta_k X_{ki}$).

As discussed by Cameron and Trivedi [61], negative binomial models are estimated via the likelihood criterion given in Expression (A7), as follows:

$$LL = \sum_{i=1}^{n} \left[Y_i \ln\left(\frac{\phi u_i}{1+\phi u_i}\right) - \frac{\ln(1+\phi u_i)}{\phi} \ln\Gamma(Y_i+1) - \ln\Gamma(\phi^{-1}) \right]$$
(A7)

which should be iterated until it reaches its maximum value.

The test for detecting overdispersion of count data proposed by Cameron and Trivedi [37] is based on Expression (A8), where H_0 is the equidispersion given by Var(Y|X) = E(Y|X), as follows:

$$Var(Y|X) = E(Y|X) + \phi[E(Y|X)]^2$$
(A8)

which is similar to the variance function of the NB model indicated by Expression (A3). For the test in Expression (A8), the significance of parameter ϕ must be verified, in which $H_1: \phi > 0$ and $H_0: \phi = 0$.

Cameron and Trivedi [37] postulated, for the detection of overdispersion in the count data and at a certain level of significance, a Poisson model should be estimated a priori. According to the authors, after this, an auxiliary ordinary least squares (OLS) model should be estimated without the intercept, whose outcome variable Y^* , given by Expression (A9), should be calculated using the fitted values of λ from the initially established Poisson model.

$$Y^* = \frac{(Y_i - \lambda_i)^2 - Y_i}{\lambda_i} \tag{A9}$$

The auxiliary model given by Expression (A10) should use λ as the sole predictor variable.

$$\widehat{Y}_i^* = \beta \lambda_i \tag{A10}$$

After the estimation of the auxiliary model in Expression (A10), Cameron and Trivedi [37] recommend checking the *p*-value from the Student's *t*-test for the predictor variable λ . In the cases where P > |t| > sig, equidispersion at a pre-established significance level is indicated; when $P > |t| \le sig$, overdispersion at a pre-established significance level is indicated. In R, the Cameron and Trivedi Test [37] can be quickly done with overdisp() command.

Appendix A.3. Zero-Inflated Poisson Regression Models

In ZIP regression models, the probability p of occurrence of a zero-count for any given observation i(i = 1, 2, ..., n), where n is the sample size), that is, $p(Y_i = 0)$, is calculated taking into account the sum of a dichotomous component with a count component and, thus, one may define the probability p_{logit_i} of occurrence of a zero-count due solely to the dichotomous component. Additionally, the probability p of occurrence of a specific count m(m = 1, 2, ...), that is, $p(Y_i = m)$, follows the expression of the Poisson probability distribution, multiplied by (p_{logit_i}) . Thus:

$$p(Y_i = m) = [p_{logit_i} + (1 - p_{logit_i}) \cdot \exp(-\lambda_i)] \mathbf{1}_{m=0} + [(1 - p_{logit_i}) \cdot \frac{\exp(-\lambda_i) \cdot \lambda_i^m}{m!}] \mathbf{1}_{m \ge 1}$$
(A11)

where $\mathbf{1}(.)$ stands for an indicator function, being $Y \sim ZIP(\lambda, p_{logit})$, where λ is the expected number of occurrences of the outcome variable for a given exposure (incidence rate ratio).

If $p_{logit_i} = 0$, clearly the probability distribution of (4) boils down to the Poisson distribution, including cases where $Y_i = 0$. In other words, the zero-inflated Poisson regression models consider two generator processes of zeros, being one due to binary distribution (the so-called structural zeros) and the other due to the Poisson distribution (the so-called sample zeros).

While the occurrence of structural zeros can be influenced by certain vector of predictor variables, the occurrence of certain *m* count can be influenced by another vector of predictor

variables. In some cases, the researcher can enter the same variable in two vectors, if the decision is to investigate whether these variable influences, concomitantly, the occurrence of the event and, if so, the amount of events (counts) of that phenomenon. Maximum likelihood method is used for obtaining the estimated values of λ and p_{logit} . From (A11), the likelihood function is given by Expression (A12):

$$L = \prod_{Y_i=0} [p_{logit_i} + (1 - p_{logit_i}) \cdot \exp(-\lambda_i)] \prod_{Y_i>0} (1 - p_{logit_i}) \cdot \lambda_i^{Y_i} \cdot \frac{\exp(-\lambda_i)}{Y_i!}$$
(A12)

Following Mouatassim et al. [62], suppose that N is the number of outcomes and N_0 is the number of zeros in the data. The likelihood becomes:

$$L = [p_{logit_i} + (1 - p_{logit_i}) \cdot \exp(-\lambda_i)]^{N_0} (1 - p_{logit_i})^{N - N_0} \prod_{m \ge 0} \lambda_i^m \cdot \frac{\exp(-\lambda_i)}{m!}$$
(A13)

Thus, the log likelihood can be written as follows:

$$LL_{1} = N_{0} \ln[p_{logit_{i}} + (1 - p_{logit_{i}}) \cdot \exp(-\lambda_{i})] + (N - N_{0})[\ln(1 - p_{logit_{i}}) - \lambda] + \ln(\lambda) \sum_{m>0} m - \sum_{m>0} \ln(m!) \quad (A14)$$

With the following equilibrium conditions at (A15) and (A16):

$$\frac{\partial}{\partial\lambda}\ln L = -N_0 \cdot \frac{(1 - p_{logit_i}) \cdot \exp(-\lambda_i)}{p_{logit_i} + (1 - p_{logit_i}) \cdot \exp(-\lambda_i)} - (N - N_0) + \frac{\sum_{m>0} m}{\lambda} = 0$$
(A15)

$$\frac{\partial}{\partial\lambda}\ln L = -N_0 \cdot \frac{(1 - p_{logit_i}) \cdot \exp(-\lambda_i)}{p_{logit_i} + (1 - p_{logit_i}) \cdot \exp(-\lambda_i)} - (N - N_0) \cdot \frac{1}{(1 - p_{logit_i})} = 0$$
(A16)

Dividing (A15) by (A16):

$$\frac{\lambda}{1 - \exp(-\lambda_i)} = \frac{\sum_{Y_i > 0} m}{N - N_0} = \frac{N\bar{m}}{N - N_0}$$
(A17)

As stated in Mouatassim et al. [62], the numerical algorithm can be used to find the estimated value of λ . Thus, the estimated value of p_{logit} can then be determined by replacing λ in (A15) or (A16).

Appendix A.4. Zero-Inflated Negative Binomial Regression Models

In ZINB regression models, the probability p of occurrence of a zero-count for any given observation i, that is, $p(Y_i = 0)$, is also calculated taking into account the sum of a dichotomous component with a count component, and the probability p of occurrence of a particular count m(m = 1, 2, ...), that is, $p(Y_i = m)$, now follows the expression of the Poisson-Gamma probability distribution:

$$p(Y_{i} = 0) = \left[p_{logit_{i}} + (1 - p_{logit_{i}}) \cdot \left(\frac{1}{1 + \phi\lambda_{i}}\right)^{\frac{1}{\phi}} \right] 1_{(m=0)} + \left\{ (1 - p_{logit_{i}}) \cdot \left(\frac{m + \phi^{-1} - 1}{\phi^{-1} - 1}\right) \cdot \left(\frac{1}{1 + \phi\lambda_{i}}\right)^{\frac{1}{\phi}} \cdot \left(\frac{\phi\lambda_{i}}{\phi\lambda_{i} + 1}\right)^{m} \right\} 1_{(m\geq1)}$$
(A18)

where **1**(.) stands for an indicator function, being now $Y \sim ZINB(\phi, \lambda, p_{logit})$ and ϕ the inverse of the shape parameter of a given Gamma distribution.

Again, if $p_{logit_i} = 0$, the probability distribution of (A18) comes down to the Poisson-Gamma distribution, including cases where $Y_i = 0$. Hence, the ZINB regression models also feature two generator processes of zeros, derived from the binary distribution and the Poisson-Gamma distribution, as stated in Cameron and Trivedi [61].

Therefore, based on (A18), one can define the log-likelihood function to estimate the parameters of a ZINB regression model as follows:

$$LL_{1} = \sum_{Y_{i}=0} \ln \left[p_{logit_{i}} + (1 - p_{logit_{i}}) \cdot \left(\frac{1}{1 + \phi\lambda_{i}}\right)^{\frac{1}{\phi}} \right] + \sum_{Y_{i}>0} \left[\ln(1 - p_{logit_{i}}) + Y_{i} \ln\left(\frac{\phi\lambda_{i}}{\phi\lambda_{i} + 1}\right) - \frac{\ln(1 + \phi\lambda_{i})}{\phi} + \ln\Gamma(Y_{i} + 1) - \ln\Gamma(\phi^{-1}) \right]$$
(A19)

The EM algorithm or the Newton-Raphson method can be used to obtain the maximum likelihood estimates.

Figure A3 shows a comparison between some Poisson, NB, ZIP and ZINB theoretical distributions.



Figure A3. Comparison between Poisson, negative binomial and zero-inflated theoretical distributions.

Appendix B

Calculated Random Effects of the Zero-Inflated Multilevel Model (ZINBM)

African Countries		
Country	Random Intercepts (v_{0j})	Random Slopes (v_{1j})
Algeria	0.29027389	0.07500865
Angola	0.36813887	0.19668883
Benin	0.94289251	0.28121811
Botswana	-0.04411956	0.02016038
Burkina Faso	-0.29742728	-0.03567919
Burundi	-0.32982355	-0.10873553
Cameroon	0.34371969	0.20256739
Central African Republic	-0.26159519	-0.03824944
Chad	0.01335540	0.00481880
Comoros	0.30241110	0.09969825
Congo (Republic of Congo)	-0.61263303	-0.29809801
Côte d'Ivoire	0.44484241	0.00888249
Djibouti	-0.66107160	-0.21794068
Eritrea	-0.87335399	0.37184683
Eswatini	-0.34670164	0.02799563
Ethiopia	0.61811679	-0.02918207
Gabon	-0.55575614	-0.22744619
Gambia	-0.50560832	-0.05546880
Ghana	-0.31605471	-0.02456802
Guinea	0.14672569	0.05072124
Guinea-Bissau	-0.01239237	-0.00206721
Kenua	-0.48414665	-0.20664528
Lesotho	-0.20847386	0.04272718
Liberia	-0.14812793	-0.13387362
Libya	-0.42687909	-0.17812012
Madagascar	-0.19426981	-0.06424881
Malawi	-0.48238576	-0.05618905
Mali	-0.03244951	-0.00550642
Mauritania	-0.42362948	0.00787552
Mauritius	-0.44565488	0 13316253
Morocco	0.59996309	-0.08021948
Mozamhiaue	0 22465744	0.06930940
Namihia	-0.56727213	0 17922600
Nioer	0.02856408	0.01123432
Nigeria	0 29337218	0 14615809
Rwanda	-0.20675524	-0.03023096
Senegal	0.47608807	0.03903200
Sierra Leone	-0.03976362	-0.01080527
South Africa	0.49076426	-0.19813893
Sudan	0.41538368	0.12070257
Tanzania	-0 14838120	-0.06700696
Тодо	-0.10477449	_0 00838254
Tunicia	0.16206770	
Luniou Haanda	_0.52724205	-0.04074073
Uzunuu Zambia	-0.52724205	-0.10404707
	0.22300120	0.000040/0
Zimhabana	0.62169026	0.29403/12
ZIMUUUWE	0.02100920	-0.0/141636

Asian Countries		
Country	Random Intercepts (v_{0j})	Random Slopes (v_{1j})
Armenia	-0.35352111	-0.09339492
Azerbaijan	0.19415641	0.09802380
Bangladesh	0.44949158	0.02262673
Bhutan	0.23931487	-0.10189279
Cambodia	-0.34036202	-0.25131552
China	-0.34968305	0.03957045
Georgia	0.00376565	0.00081100
India	0.49470753	-0.04487317
Indonesia	0.55611730	0.25351292
Japan	-0.47067235	0.33723226
Kazakhstan	0.06968353	0.02646168
Kyrgyzstan	0.16421008	0.04128285
Laos	-0.36615737	-0.09463578
Malaysia	-0.05021094	0.02749636
Mongolia	-0.30501764	0.00854783
Nepal	-0.12802811	-0.02303344
Philippines	0.06464324	-0.00243171
Singapore	0.05540805	-0.06245646
South Korea	-0.27408919	0.06821146
Sri Lanka	-0.11471574	0.00593398
Tajikistan	-0.42844117	-0.25478445
Thailand	0.80785575	-0.02971311
Turkmenistan	-0.50514492	-0.30576730
Uzbekistan	-0.35409601	-0.16818937
Vietnam	-0.12923459	-0.02394848

Middle	East	Countries
--------	------	-----------

Country	Random Intercepts (v_{0j})	Random Slopes (v_{1j})
Bahrain	0.12970880	-0.05181181
Cyprus	-0.48447181	0.38519245
Egypt	0.91216283	-0.04196600
Iran	-0.39907911	-0.08266890
Israel	-0.23540525	0.18956769
Jordan	-0.69146088	0.20780363
Kuwait	0.91731159	-0.62780406
Lebanon	-0.79813114	0.00007300
Oman	-0.21215604	0.13006397
Pakistan	0.44354709	0.13218649
Saudi Arabia	0.22240830	-0.08234248
Syria	0.41615624	0.07051341
Turkey	-0.11882707	0.02221109
United Arab Emirates	-0.24431332	0.13840813
Yemen	-0.57927546	-0.09647853

European Countries			
Country	Random Intercepts (v_{0j})	Random Slopes (v_{1j})	
Albania	0.44277582	0.18957083	
Austria	0.45259684	-0.45199415	
Belarus	-0.57872150	-0.10517272	
Belgium	0.00566068	-0.00420827	
Bosnia & Herzegovina	0.27446995	0.00448070	
Bulgaria	1.12489858	0.12753296	
Croatia	-0.18674269	-0.00045776	
Czechia	0.18708253	-0.07003003	
Denmark	-0.12906556	0.14799282	
Estonia	-0.05015729	0.02194181	
Finland	-0.50544598	0.57713222	
France	0.24695152	-0.22664305	
Germany	-0.03254498	0.03421383	
Greece	0.04097844	-0.02443984	
Hungary	-0.27064047	0.14312356	
Ireland	-0.00913212	0.00946188	
Italy	0.76640023	-0.50200084	
Latvia	-0.30630650	0.04069200	
Lithuania	-0.61513082	0.14220830	
Moldova	-0.75627015	-0.09149125	
Netherlands	-0.03288228	0.03697565	
North Macedonia	-0.51077076	0.00863490	
Norway	-0.23699016	0.25795913	
Poland	-0.45472045	0.19941508	
Portugal	0.67046682	-0.57361285	
Romania	-0.04864107	-0.00168571	
Russia	-0.84761028	-0.21218648	
Slovakia	-0.06970213	0.01001509	
Slovenia	0.26469128	-0.15521686	
Spain	0.58641231	-0.50766076	
Sweden	-0.23326544	0.26606757	
Switzerland	-0.59286471	0.68130772	
Ukraine	0.36722275	0.14899188	
United Kingdom	-0.54808365	0.59353453	
Yugoslavia	0.08347056	0.03895000	

North American and Caribbean Countries			
Country	Random Intercepts (v_{0j})	Random Slopes (v_{1j})	
Canada	-0.17267838	0.19520823	
Dominican Republic	-0.58287177	-0.07829544	
Haiti	0.20322620	0.07527424	
Jamaica	-0.24794002	0.00938118	
Trinidad & Tobago	0.37387195	-0.09694391	

Oceania Countries			
Country	Random Intercepts (v_{0j})	Random Slopes (v_{1j})	
Australia	-0.10445492	0.10975066	
Fiji	0.02271001	-0.00678572	
New Zealand	-0.57624262	0.65679904	
Papua New Guinea	0.27204064	0.07077443	

South American Countries

Country	Random Intercepts (v_{0j})	Random Slopes (v_{1j})
Argentina	0.31771912	-0.01948855
Bolivia	-0.44415219	-0.02577823
Brazil	0.96602883	-0.23480633
Chile	0.66125855	-0.48323508
Colombia	-0.29419744	-0.05614823
Costa Rica	0.40459963	-0.21728966
Ecuador	-0.30005928	-0.08591258
El Salvador	-0.02803395	0.00091081
Guatemala	-0.58738992	-0.12267548
Guyana	-0.39814087	0.01490841
Honduras	-0.38180429	-0.11161159
Mexico	-0.35404205	-0.01542422
Nicaragua	0.16002692	0.04744322
Panama	-0.30726926	0.00814645
Paraguay	0.13283151	0.06199097
Peru	0.16886576	-0.01576054
Uruguay	-0.04140083	0.01679624
Venezuela	-0.02358928	-0.00732867

References

- Blevins, D.P.; Tsang, E.W.K.; Spain, S.M. Count-Based Research in Management: Suggestions for improvement. Organ. Res. Methods 2015, 18, 47–69. [CrossRef]
- 2. Almeida, B.P.; Gonçalves, E.; da Silva, A.S.; Reis, R. Internalization of Knowledge Spillovers by Regions: A measure based on self-citation patents. *Ann. Reg. Sci.* 2020, *66*, 309–330. [CrossRef]
- 3. O'Hara, R.; Kotze, D.J. Do Not Log-Transform Count Data. Nat. Preced. 2010, 1, 188–122.
- 4. Lambert, D. Zero-Inflated Poisson Regression, with an Application to Defects in Manufacturing. *Technometrics* **1992**, *34*, 1–14. [CrossRef]
- 5. Spriensma, A.S.; Hajos, T.R.S.; de Boer, M.R.; Heymans, M.W.; Twisk, J.W.R. A New Approach to Analyse Longitudinal Epidemiological Data with an Excess of Zeros. *BMC Med. Res. Methodol.* **2013**, *1*, 13–27. [CrossRef] [PubMed]
- 6. Heck, R.; Thomas, S.L. *An Introduction to Multilevel Modeling Techniques: MLM and SEM Approaches Using Mplus*, 3rd ed.; Routledge: New York, NY, USA, 2015; p. 461.
- 7. Mathieu, J.E.; Chen, G. The Etiology of the Multilevel Paradigm in Management Research. J. Manag. 2011, 37, 610–641. [CrossRef]
- 8. Courgeau, D. Methodology and Epistemology of Multilevel Analysis: Approaches from different Social Sciences; Springer: Paris, France, 2012; p. 240.
- 9. Arceneaux, K.; Nickerson, D.W. Modeling Certainty with Clustered Data: A comparison of methods. *Political Anal.* 2009, 17, 177–190. [CrossRef]
- Hall, D.B. Zero-Inflated Poisson and Binomial Regression with Random Effects: A case study. *Biometrics* 2000, 56, 1030–1039. [CrossRef] [PubMed]
- 11. Fisman, R.; Miguel, E. Corruption, Norms, and Legal Enforcement: Evidence from diplomatic parking tickets. J. Polit. Econ. 2007, 115, 1020–1048. [CrossRef]
- 12. Shook, C.L.; Ketchen, D.J.; Cycyota, C.S.; Crockett, D. Data Analytic Trends and Training in Strategic Management. *Strat. Mgmt. J.* **2003**, *24*, 1231–1237. [CrossRef]

- 13. Perumean-Chaney, S.; Morgan, C.J.; McDowall, D.; Aban, A.B. Zero-Inflated and Overdispersed: What's one to do? *J. Stat. Comput. Simul.* **2013**, *83*, 1671–1683. [CrossRef]
- 14. Pew, T.; Warr, R.L.; McDSchultz, G.G.; Heaton, M. Justification for Considering Zero-Inflated Models in Crash Frequency Analysis. *Transp. Res. Interdiscip. Perspect.* 2020, *8*, 1671–1683. [CrossRef]
- 15. Lee, S. Addressing Imbalanced Insurance Data Through Zero-Inflated Poisson Regression with Boosting. *ASTIN Bull.* 2021, *51*, 27–55. [CrossRef]
- 16. Diaz, M.; Huff-Corzine, L.; Corzine, J. Demanding Reduction: A County-level analysis examining structural determinants of human trafficking arrests in Florida. *Crime Deling*. **2020**, 1–24. [CrossRef]
- 17. Koning, I.; de Looze, M.; Harakeh, Z. Parental Alcohol-Specific Rules Effectively Reduce Adolescents' Tobacco and Cannabis Use: A longitudinal study. *Drug Alcohol Depend.* **2020**, *216*, 1–6. [CrossRef]
- Chinaeke, E.; Melanie, G.; Yuan, H.; Jiajia, Z.; Bankole, O. Parental The Positive Association Between Employment and Self-Reported Mental Health in the USA: A robust application of marginalized zero-inflated negative binomial regression (MZINB). J. Public Health 2020, 42, 340–352.
- 19. Clouston, S.A.P.; Natale, G.; Link, B.J. Socioeconomic Inequalities in the Spread of Coronavirus-19 in the United States: A examination of the emergence of social inequalities. *Soc. Sci. Med.* **2021**, *268*, 113554. [CrossRef]
- 20. Karmakar, M.; Lantz, P.M.; Tipirneni, R. Association of Social and Demographic Factors With COVID-19 Incidence and Death Rates in the US. *JAMA* **2021**, *4*, e2036462.
- 21. Bolker, B.M. Linear and Generalized Linear Mixed Models. In *Ecological Statistics: Contemporary Theory and Application;* Fox, G.A., Negrete-Yankelevich, S., Sosa, V.J., Eds.; Oxford University Press: Oxford, UK, 2015; p. 406.
- 22. Brooks, M.E.; Kristensen, K.; van Benthem, K.J.; Magnusson, A.; Berg, C.W.; Nielsen, A.; Skaug, H.J.; Mächler, M.; Bolker, B.M. glmmTMB Balances Speed and Flexibility Among Packages for Zero-inflated Generalized Linear Mixed Modeling. *R J.* **2017**, *9*, 378–400. [CrossRef]
- 23. Woltman, H.; Feldstain, A.; MacKay, J.C.; Rocchi, M. An Introduction to Hierarchical Linear Modeling. *TQMP* **2012**, *8*, 52–69. [CrossRef]
- 24. DeBruine, L.M.; Barr, D.J. Understanding Mixed-Effects Models Through Data Simulation. AMPPS 2021, 4, 1–15.
- 25. Hair, J.F., Jr.; Fávero, L.P. Multilevel Modeling for Longitudinal Data: Concepts and applications. *Rausp Manag. J.* 2010, 54, 459–489. [CrossRef]
- 26. Meteyard, L.; Davies, R.A.I. Best Practice Guidance for Linear Mixed-Effects Models in Psychological Science. *J. Mem. Lang.* 2020, 112, 104092. [CrossRef]
- 27. Parker, R.A.; Scott, C.; Inácio, V.; Stevens, N.T. Using Multiple Agreement Methods for Continuous Repeated Measures Data: A tutorial for practitioners. *BMC Med. Res. Methodol.* **2020**, *20*, 154. [CrossRef]
- 28. Hox, J. Multilevel Analysis: Techniques and Applications, 3rd ed.; Routledge: New York, NY, USA, 2017; p. 364.
- Fávero, L.P.; Belfiore, P. Data Science for Business and Decision Making; Academic Press Elsevier: Cambridge, MA, USA, 2019; p. 1244.
- 30. Finch, W.H.; Bolin, J.E.; Kelley, K. Multilevel Modeling Using R, 2nd ed.; Chapman and Hall: New York, NY, USA, 2019; p. 254.
- 31. Garson, G. *Multilevel Modeling: Applications in STATA*[®], *IBM*[®] *SPSS*[®], *SAS*[®], *R*, & *HLM*[™]; Sage Publications: Thousand Oaks, CA, USA, 2019; p. 552.
- 32. Nelder, J.A.; Wedderburn, W.M. Generalized Linear Models. J. R. Stat. Soc. 1972, 135, 370–384. [CrossRef]
- Cameron, A.C.; Trivedi, P.K. Regression Analysis of Count Data, 2nd ed.; Cambridge University Press: Cambridge, MA, USA, 2013; p. 597.
- 34. Hilbe, J.M. Negative Binomial Regression, 2nd ed.; Cambridge University Press: Cambridge, MA, USA, 2011; p. 576.
- 35. Hilbe, J.M. *Modeling Count Data*; Cambridge University Press: Cambridge, MA, USA, 2014; p. 300.
- 36. Wooldridge, J.M. Econometric Analysis of Cross Section and Panel Data, 2nd ed.; MIT Press: Cambridge, MA, USA, 2010; p. 1095.
- 37. Cameron, A.C.; Trivedi, P.K. Regression-Based Tests for Overdispersion in the Poisson Model. J. Econ. 1990, 46, 347–364. [CrossRef]
- 38. Fávero, L.P.; Belfiore, P.; Santos, M.A.; Freitas Souza, R. A Stata (and Mata) Package for Direct Detection of Overdispersion in Poisson and Negative Binomial Regression Models. *Stat. Optim. Inf. Comput.* **2020**, *8*, 773–789. [CrossRef]
- Payne, E.H.; Hardin, J.W.; Egede, L.E.; Ramakrishnan, V.; Selassie, A.; Gebregziabher, M. Approaches for Dealing with Various Sources of Overdispersion in Modeling Count Data: Scale adjustment versus modeling. *Stat. Methods Med. Res.* 2017, 26, 1802–1823. [CrossRef]
- 40. Fávero, L.P.; Serra, R.G.; Santos, M.A.; Brunaldi, E. Cross-Classified Multilevel Determinants of Firm's Sales Growth in Latin America. *IJOEM* 2018, *13*, 902–924. [CrossRef]
- 41. Ngatchou-Wandji, J.; Paris, C. On the Zero-Inflated Count Models with Application to Modelling Annual Trends in Incidences of Some Occupational Allergic Diseases in France. *J. Data Sci.* **2011**, *69*, 639–659.
- 42. Vuong, Q.H. Likelihood Ratio Tests for Model Selection and Non-Nested Hypotheses. Econometrica 1989, 57, 307–333. [CrossRef]
- 43. Goldstein, H. Multilevel Statistical Models, 4th ed.; Wiley Series in Probability and Statistics; Wiley: Chichester, UK, 2010; p. 384.
- 44. Santos, M.A.; Fávero, L.P.; Distadio, L.F. Adoption of the International Financial Reporting Standards (IFRS) on Companies' Financing Structure in Emerging Economies. *Financ. Res. Lett.* **2016**, *16*, 179–189. [CrossRef]

- 45. Serra, R.G.; Fávero, L.P. Multiples' Valuation: The Selection of Cross-Border Comparable Firms. *Emerg. Mark. Financ. Trade* 2018, 54, 1973–1992. [CrossRef]
- 46. Fávero, L.P. The Zero-Inflated Negative Binomial Multilevel Model: Demonstrated by a Brazilian dataset. *IJMOR* **2018**, *11*, 90–107. [CrossRef]
- 47. Lee, A.H.; Wang, K.; Scott, J.A.; Yau, K.K.W.; McLachlan, G.J. Multilevel Zero-Inflated Poisson Regression Modelling of Correlated Count Data with Excess Zeros. *Stat. Methods Med. Res.* **2006**, *15*, 47–61. [CrossRef]
- 48. Rabe-Hesketh, S.; Skrondal, A.; Pickels, A. Maximum Likelihood Estimation of Limited and Discrete Dependent Variable Models with Nested Random Effects. *J. Econ.* 2005, *128*, 301–323. [CrossRef]
- 49. Mauro, P. Corruption and Growth. Q. J. Econ. 1995, 110, 681–712. [CrossRef]
- Duggan, M.; Levitt, S.D. Winning Isn't Everything: Corruption in Sumo Wrestling. *Am. Econ. Rev.* 2002, *92*, 1594–1605. [CrossRef]
 Glaeser, E.L.; Goldin, C. *Corruption and Reform: Lessons from America's Economic History*; University of Chicago Press: Chigago, IL,
- USA, 2006; p. 384.
 52. Levitt, S.D. White-Collar Crime Writ Small: A case study of bagels, donuts, and the honor system. *Am. Econ. Rev.* 2006, *96*, 290–294. [CrossRef]
- 53. Svensson, J. Eight Questions about Corruption. J. Econ. Perspect. 2005, 19, 19–42. [CrossRef]
- 54. Desmarais, B.A.; Harden, J.J. Testing for Zero Inflation in Count Models: Bias Correction for the Vuong Test. *Stata J.* **2013**, *13*, 810–835. [CrossRef]
- 55. Okhuysen, G.; Bonardi, J.-P. The Challenges of Building Theory by Combining Lenses. AMR 2011, 36, 6–11. [CrossRef]
- 56. Bettis, R.; Gambardella, A.; Helfat, C.; Mitchell, W. Quantitative empirical analysis in strategic management: Editorial. *Strateg. Manag. J.* **2014**, *35*, 949–953. [CrossRef]
- 57. Dale, D.; Sirchenko, A. Estimation of Nested and Zero-Inflated Ordered Probit Models. Stata J. 2021, 21, 3–38. [CrossRef]
- 58. Antonakis, J.; Bastardoz, N.; Rönkkö, M. On Ignoring the Random Effects Assumption in Multilevel Models: Review, critique, and recommendations. *Organ. Res. Methods* **2021**, *24*, 443–483. [CrossRef]
- 59. Klakattawi, H.; Vinciotti, V.; Yu, K. A Simple and Adaptive Dispersion Regression Model for Count Data. *Entropy* **2018**, *20*, 142. [CrossRef]
- 60. Favero, L.P.; Santos, M.A.; Serra, R.G. Cross-Border Branching in the Latin American Banking Sector. *IJBM* **2018**, *4*, 496–528. [CrossRef]
- 61. Cameron, A.C.; Trivedi, P.K. Microeconomics using Stata; Stata Press: College Station, TX, USA, 2010; p. 706.
- 62. Mouatassim, Y.; Ezzahid, E.H.; Belasri, Y. Operational Value-at-Risk in Case of Zero-Inflated Frequency. *IJEF* 2012, *4*, 70–77. [CrossRef]